3pack

6pack
to PPF0.

should follow the same route as the tapes and be continuous

Screens need not envelop the cables within the tracker but

Tapes are stackes in threes with a foil screen for each.

Proposed grounding scheme for Atlas Forward SCT      Tony smith September ’01
between the detectors and the shield and

Carbon fibre cylinders
screens of all conventional cables DC isolated

Power supplies
in conductors which returns via cable screens
switched mode supplies can generate current flow

Parasitic common mode voltage generator from

Vannular Al foil.

foils between PPF0s.

Cooling tubes are
annular foil
cooling tube and optional

PPF0 position connect to

Radial foils from each
except for single point ground connection to minimise
current flow in the shield.

Thermal shield is DC isolated from other detectors

PPF2

currents
should be capable of handling large fault
and that of the annular conductor used
shield. The cross section of this conductor

PPF1 , PPF2 or at the end of the thermal
made to any of the annular conductors at
and hence ground of other detectors is

Single point tie to experiment ground
and PPF2

 and metalwork together at PPF1
cable screens, cooling tubes

Annular conductor connects all
control room ground
experiment ground
edge of the disk at multiple points.

connected to the annular tape around the
shield. A conductive foil should be placed
proximity to the inner face of the thermal

Most forward disk has modules in close
cooling tubes, heat spreaders and tape shields

Equipotential bonding of all conducting structures
common mode return current in shields.

power supply rather than at patch panels reduces
common mode filter incorporated into
current in cable
forms a true common mode choke to reduce HF
split torroidal ferrite clipped around cable
in cable screens due to mag fields in long runs
from each other except at PPf2 to avoid current flow
back to back zeners
for safety.

across this if needed
sheathed cable enters
to avoid LF current flow
connected to chassis

Cable shield is AC

PSU on a pin
from cover and enters
shield is isolated
connector cover but

Rack grounds tied to

Outer thermal enclosure
tape foils shorted to enclosure
at their entry point
shorted to PPF0 tape foils
all tape foil screens shorted
at PPF0 with an annular Al
tape.

Carbon fibre disk shorted
with Al foil to annular
foils between PPF0s

Module mounting points
connected together
electrically by
cooling tubes and
optionally augmented
with annular Al foil
shorted to the annular
screens and structure
the module power returns and the cable

This forms the only DC connection between
together by cooling tubes and optional
cooling blocks which are shorted

(ground planes) are connected to

Module signal reference points
to cooling plant
before diverging
in cooling tubes

Isolating section
rack grounds
crate grounds tied to
tied to crate ground

Power supply front panels

PPF3

PPF1

Al power tapes

Thin conventional cable

Thick conventional cable

PPF0